Author’s response to reviews

Title: Design and clinical validation of a software program for automated measurement of mammographic breast density

Authors:
Adriano Araujo (adrianorad@gmail.com)
HELIANA SOARES (helianabs@gmail.com)
DANIEL CARVALHO (danielfasanaro@gmail.com)
ROBERTO MENDONÇA (robertomorenomendonca@gmail.com)
ANTONIO OLIVEIRA (oliveira.amg@gmail.com)

Version: 2 Date: 20 Feb 2020

Author’s response to reviews:
We try to respond to requests from reviewers and are available for any additional changes.

Response letter:
1 - Please be sure to assign your manuscript to a Section/Category upon re-submission.

The section / category of the manuscript was designated

2 - Please remove any files from the file inventory that you do not wish to see published.

Files that would not be published were removed.

3 - Please move your Declarations to follow the Abbreviations.

This change was made
4 - The 'Consent for publication' section covers manuscripts that contain any individual person’s data in any form (including individual details, images or videos), for which consent to publish must be obtained from that person, or in the case of children, their parent or legal guardian. If your manuscript does not contain any individual person’s data, please state “Not applicable” in this section.

Our article contains personal data, such as weight, height and breast density, and, therefore, informed consent was obtained.

5 - At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text. Should you wish to respond to these revision requests, please include the information in the designated input box only.

In our view the article is a clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.